



Research Centre for International Economics

Working Paper: 2013050

Title of Paper

Heterogeneous Background Risks and Portfolio Choice: Evidence from Micro-Level Data

Authors' List

Darius Palia, Rutgers University Yaxuan Qi, City University of Hong Kong Yangru Wu, Rutgers University

Abstract

We construct a set of household-level background risk variables to capture the covariance structure of three nonfinancial assets and two financial assets. These risks are in general statistically significant and economically important for a household's stock market participation and stock holdings. A one standard deviation increase in background risks reduces the participation probability by 11% and the stock holdings to wealth ratio by 4%. The volatilities of labor income, housing value, and business income reduce a household's participation and stock holdings. A household with labor income highly correlated with stock (bond) returns is less (more) likely to invest in stock.

© 2013 by Yaxuan Qi. All rights reserved. Short sections of text, not to exceed two paragraphs, may be quoted without explicit permission provided that full credit, including © notice, is given to the source.

Heterogeneous Background Risks and Portfolio Choice: Evidence from Micro-Level Data

Darius Palia, Yaxuan Qi, and Yangru Wu Final Version: October 6, 2013

Abstract We construct a set of household-level background risk variables to capture the covariance structure of three nonfinancial assets and two financial assets. These risks are in general statistically significant and economically important for a household's stock market participation and stock holdings. A one standard deviation increase in background risks reduces the participation probability by 11% and the stock holdings to wealth ratio by 4%. The volatilities of labor income, housing value, and business income reduce a household's participation and stock holdings. A household with labor income highly correlated with stock (bond) returns is less (more) likely to invest in stock.

Key words: background risks, stock market participation, portfolio choice *JEL Codes*: G11, G12

Darius Palia is a Professor at the Rutgers Business School, Rutgers University, dpalia@business.rutgers.edu; Yaxuan Qi is an Associate Professor at the Department of Economics and Finance, City University of Hong Kong, yaxuangi@cityu.edu.hk; and Yangru Wu is a Professor at the Rutgers Business School, Rutgers University and a Visiting Professor at the Chinese Academy of Finance and Development, Central University of Finance and Economics, yangruwu@business.rutgers.edu. We thank Douglas Breeden, Ivan Brick, John Campbell, John Cochrane, Mariano Croce, Stephen Dimmock, Lee Dunham, Esther Eiling, William Greene, Harrison Hong, Ravi Jagannathan, Raymond Kan, Ron Kaniel, Simi Kedia, Hong Liu, Deborah Lucas, Evgeny Lyandres, Imants Paeglis, Ben Sopranzetti, Annette Vissing-Jorgensen, John Wald, Akiko Watanable, Amir Yaron, Motohiro Yogo, Feng Zhao, the Editor (Kenneth West), two anonymous referees and seminar participants at the Central University of Finance and Economics, Concordia, Northwestern, Rutgers, Texas at San Antonio, Toronto, the FMA conference, the Duke/UNC Asset Pricing Conference, the International Symposium on Financial Engineering and Risk Management, the Singapore International Conference on Finance, the Northern Finance Association conference, and the American Economic Association meetings for helpful comments and discussions. Palia and Wu thank the Whitcomb Financial Services Center for partial financial support. Qi thanks financial support from the Institut de Finance Mathématique de Montréal (IFM2) and Fonds de recherche du Québec Société et culture (FQRSC). Part of this work was completed while Wu visited the Central University of Finance and Economics whose hospitality was greatly acknowledged. All errors remain our own responsibility. Corresponding author: Yangru Wu, yangruwu@business.rutgers.edu.

This paper examines how risks sourced from non-tradable/illiquid assets, such as labor, housing, and private business, affect stock market participation and portfolio allocation decisions of a household. Following Heaton and Lucas (2000a), we term these non-financial market risks as background risks. Campbell (2006), in his AFA Presidential Address, advocates the importance of the existence of non-tradable assets (human capital) and illiquid assets (owneroccupied house) in determining a household's asset allocation.¹ Standard asset pricing theory suggests that in complete markets, background risks should have no influence on an investor's portfolio choice because these risks can be fully insured by trading financial securities. However, when markets are incomplete such that these risks are not entirely spanned by financial assets, a household will alter its portfolio to offset its idiosyncratic background risks (e.g., Duffie et al., 1997; Constantinides and Duffie, 1996; Heaton and Lucas, 1996 and 2000b; Viceira 2001; and Cochrane, 2008). Consequently, a household's optimal portfolio depends on its exposure to background risks. This paper aims to provide some insight on whether and how the heterogeneity of background risks across households can help explain the large fraction of non-stockholders, i.e., the limited stock market participation puzzle (Mankiw and Zeldes, 1991) and the enormous cross-sectional variation in households' stock holdings.

The importance of background risks on asset allocation has received considerable attention in the financial economics literature. While numerous papers have studied this topic, there is still much disagreement on whether the existence of heterogeneous background risks can help explain the observed variation of stock investments among households. Theoretical models and numerical simulation studies are sensitive to the assumptions on the properties of non-financial income/assets (Heaton and Lucas, 1996, 1997, and 2000a; Haliassos and Michaelides, 2003; Cocco *et al*, 2005; Benzoni *et al*, 2007; Storesletten *et al*, 2004; and Krueger and Lustig,

2010). Research using micro-level data is in paucity, partly due to the difficulty of estimating background risks at the household level. Prior studies yield mixed results, probably due to the difference of selected samples (Haliassos and Bertaut, 1995; Vissing-jorgensen, 2002; Heaton and Lucas, 2000b; Massa and Simonov, 2006; and Angeres and Lam, 2009). This paper uses a long panel of a large sample of U.S. households to study the impact of three non-tradable/illiquid assets, namely, labor, housing and private business, on a household's stock investment decisions. To the best of our knowledge, our paper is the first to comprehensively examine these three backgrounds risks, which are advocated and studied separately in the prior literature.

Motivated by the mean-variance analysis (e.g., Davis and Willen, 2002; Flavin and Yamashita 2002; and Cochrane, 2008), we characterize the *variance-covariance structure* generated by the three non-financial assets and two financial assets. Specifically, we use the annual growth rates of labor income, home equity, and business income to proxy returns from human capital, housing, and private business, respectively. For each household, we estimate the standard deviations of these growth rates. We further calculate the correlations of these growth rates with stock returns and with the risk-free rate. We then use a Logit regression to examine how these background risk variables impact a household's stock market participation, and a Tobit regression to study their effects on a household's stock holdings.² We extend the empirical literature on the importance of background risks in a household's portfolio decision in the following ways:

First, by jointly studying the three types of background risks, we are able to quantitatively evaluate their relative importance. We show that all three types of background risks are in general statistically significant and economically important. The existence of labor income or owner-occupied house encourages stock investment whereas the existence of private business reduces stock investment. If all the background risk variables shift one standard deviation from their sample means, the probability of stock participation decreases by 10.82 percent and the stock holding proportion to financial wealth drops by 3.69 percent.³ Given that the sample average of stock participation is 38.8 percent and the average stock holding proportion to financial wealth is 20.7 percent, we argue that the economic impact of background risks on the cross-sectional variation of portfolio choice is substantial. In terms of relative importance, labor income is the most important, followed by housing, while the impact of business income is limited.⁴

Second, we find strong evidence in support of the hedging motive (e.g., Viceira, 2001), suggesting that investors alter their portfolios to hedge their labor income risk. The volatility of labor income growth significantly reduces stock market participation and the proportion of wealth invested in stocks, consistent with the notion that risky labor income reduces investment in risky assets. Moreover, a household with stock-like labor income (i.e., labor income is highly correlated with stock returns) is less likely to participate in stock market and allocates a smaller portion of wealth to stocks. In contrast, a household with riskfree-like labor income (i.e., labor income is highly correlated with the risk-free asset) is more likely to participate in stock market and invests more wealth in stocks. These findings help reconcile the contradicting results provided by prior numerical simulations. For example, Heaton and Lucas (1996) argue that the inclusion of labor income risks is in general unable to explain the observed low stock investment by households. Their model assumes a low correlation between labor income and stock returns and therefore labor income works like a safe asset which stimulates stock investment. Benzoni et al. (2007) assume that labor income and stock dividends are cointegrated, and thus labor income and stock returns are highly correlated in the long run. Consequently, the stock-like labor income

reduces stock investment. In addition, our findings are different from Massa and Simonov (2006), who find that households tend to hold stocks that are geographically and professionally close to them. Their finding is against the hedging motive but is in favor of the familiarity and home bias hypothesis. Our findings support the hedging motive and are robust after controlling for industry and state fixed effects.

Third, we examine the impact of owner-occupied housing on stock investment. We find that the volatility of home equity growth significantly reduces stock market participation and stock holdings. Owner-occupied housing functions more like "bond-like" asset. The correlation of home equity growth with the risk-free rate has a positive impact on participation and stock holdings, while the correlation of home equity with stock returns has no significant impact. This finding echoes the notion that housing investment is a good hedge of inflation risk (Goetzmann and Valaitis, 2006). To our knowledge, this is the first paper that uses household-level data to directly estimate the impact of correlation between housing value and financial assets on stock investment. Our finding compliments the studies advocating the importance of owner-occupied housing on stock investment (Flavin and Yamashita, 2002; Cocco, 2005; and Yao and Zhang 2005).

Fourth, we examine the interactive effect of education and background risks on stock participation and stock holdings. Mankiw and Zeldes (1991) and Vissing-Jorgensen (2002), among others, suggest that education is a proxy for transaction costs and find that education has a significant impact on a household's portfolio decision. We extend this research by finding that the change of background risks has a more pronounced effect on more highly educated households. Specifically, when all background risk variables increase by one standard deviation from their sample means, a household whose head has a college degree will decrease its likelihood to participate in the stock market by 12.53 percent and reduce its proportion of stock holdings by 4.24 percent, whereas a household whose head has no high school education will decrease its likelihood of participation by only 8.43 percent and reduce its stock holdings by 3.07 percent.

Overall, we empirically demonstrate the importance of background risks in determining a household's portfolio choice. We document the enormous variation of background risks across households, and argue that this heterogeneity helps explain the limited stock market participation puzzle and the observed cross-sectional variation in stock holdings.

The remainder of this paper is organized as follows. Section 1 reviews the literature and describes our empirical design. Section 2 details the data with summary statistics. Section 3 presents the empirical results and Section 4 concludes.

1. RELATED LITERATURE AND EMPIRICAL MODEL

In this section, we first discuss the construction of background risk factors. Based on theoretical studies in the prior literature, we generate testing hypotheses about the predicted impacts of background risk factors on portfolio choice. We then specify our empirical model and discuss econometric issues and measurement errors. The definition of our background risk variables and their predicted (expected) impacts on stock investment are summarized in Table 1.

1.1 Background Risks Measures

We aim to develop an empirical model which allows us to examine how the heterogeneity of background risk exposures among households affects the cross-sectional variation of stock investment. More importantly, we want to jointly consider three types of nonfinancial assets - labor, housing and private business. While each of these background risks has been separately investigated in the literature, to the best of our knowledge, no study has jointly considered all three types of risks in an optimizing model. Developing a theoretical model of optimal portfolio choice in the presence of the three types of background risks is beyond the scope of this paper. We therefore borrow the insights from the prior literature to build a reducedform empirical model to test the predicted impacts of these risks on household stock investment. We aim to examine whether these background risks are quantitatively important to a household's stock investment. Specifically, we investigate whether the cross-household variation of background risks can help explain the large fraction of non-stock investment, and the observed enormous cross-sectional variation in households' stock holdings.

We consider an economy with three types of non-financial assets: labor, housing, and private business, and two financial assets: risky stock and risk-free bond. Motivated by the standard mean-variance framework, we argue that the optimal portfolio is determined by the variance-covariance structure of returns of these assets. Assuming that the investor cannot trade non-financial assets, optimal portfolio choice involves selecting a combination of stock and riskfree asset to minimize the overall risk exposure to all risky assets A household would choose a less risky portfolio if it is exposed to more unfavorable background risks. Assuming that shortsale is prohibited, zero-stockholding (i.e., non-stock market participation) can yield as an optimal choice if a household is highly exposed to background risks. Therefore, a household exposed to more unfavorable background risks is expected to allocate less wealth to stocks or is less likely to participate in the stock market.

This hypothesis is motivated by a large body of literature, which provides theoretical predictions on how background risk affects portfolio choice. For example, Constantinides and

Duffie (1996), and Viceira (2001) suggest that investors alter stock investment to hedge labor income risk. Cocco (2005), Yao and Zhang (2005), and Flavin and Yamashita (2002) show that the existence of risky housing reduces stock investment. Empirical studies based on numerical calibration show that the existence of background risks cannot fully explain the enormous variation of stock holdings across households, especially the large fraction of non-stock holding (e.g., Heaton and Lucas, 1996, 1997, and 2000a; Haliassos and Michaelides, 2003; and Cocco et al, 2005). Research using micro-level data in general confirms the importance of background risks but yields mixed results regarding how background risks affect portfolio choice. Haliassos and Bertaut (1995) demonstrate that investors in high-risk occupations hold less stock. Vissing-Jorgensen (2002) finds that a larger standard deviation of non-financial income reduces stock investment, but the covariance of income and stock returns has no impact. Heaton and Lucas (2000b) show that investors invest less in stocks when they face more volatile business income, but labor income risk does not significantly affect stock investment. Angerer and Lam (2009) report that a permanent income shock reduces stock investment but a transitory income shock does not. Guiso et al. (1996) and Hochguetel (2002), respectively, use data from Italy and the Netherlands to show that households exposed to higher labor income risk hold safer portfolios. Chen et al. (2007) and Dimmock (2012) argue that background risks also affect asset allocation of institutional investors. Eiling (2013) uses the industry-level labor income to show that human capital affects the cross-sectional stock returns.

Our approach is to construct a set of factors that capture the "*unfavorable*" background risk exposures. Motivated by the mean-variance framework, we consider the covariance of returns between financial and non-financial assets. One big challenge in this literature is that returns of these non-financial assets are not observable. Following Jagannathan and Wang (1996), Flavin and Yamashita (2002) and Heaton and Lucas (2000b), we use annual growth rates of labor income, home equity, and business income to proxy their respective returns.⁵ We then estimate the covariance between these annual growth rates and returns of financial assets. Our baseline model consists of 12 background risk factors, which can be grouped into four categories.

First, we consider the standard deviations of growth rates of labor income, home equity, and business income, denoted by Std(Lab), Std(Hou) and Std(Bus), respectively. Previous research shows that the volatility of additional risky income reduces the demand for stock (Heaton and Lucas, 2000a, b; Vissing-Jorgensen, 2002; Hochguertel, 2002; Guiso *et al.*, 1996; and Angerer and Lam, 2009). Hence, we expect each of these variables to have a negative effect on the proportion of stock holdings and on stock market participation.

Second, we calculate the correlations between the three growth rates and stock returns, denoted by $Corr(R_s, Lab)$, $Corr(R_s, Hou)$, and $Corr(R_s, Bus)$, respectively. The correlation between a background risk shock and stock returns is potentially important to a household's portfolio choice (Viceira, 2001; and Benzoni *et al.*, 2007). For example, a positive correlation between labor income and stock returns reduces a household's willingness to hold stock because labor income substitutes for stock. On the other hand, a negative correlation between labor income and stock returns encourages stock holdings because stock can be used as a hedge against labor income risk. We hence expect $Corr(R_s, Lab)$, $Corr(R_s, Hou)$, and $Corr(R_s, Bus)$ to carry negative coefficients.

Third, we include correlations between the three growth rates and the risk-free rate, denoted by $Corr(R_f, Lab)$, $Corr(R_f, Hou)$ and $Corr(R_f, Bus)$, respectively. We expect each of these variables to have a positive effect on stock market participation and stock holdings. The measure introduced here captures the co-movement of a background risk variable and the real interest rate, which is primarily driven by unexpected inflation. This design is to test whether bond-like income reduces the pressure on precautionary savings, whereby encouraging investment in stocks (e.g., Cocco *et al.*, 2005). Intuitively, a household with stable labor income which increases with inflation (for example, those working in the public sector) is more likely to invest in risky stocks because its labor income risk is lower.

Fourth, we include the correlations of the returns among the three non-financial assets, denoted by *Corr(Lab,Hou)*, *Corr(Lab,Bus)* and *Corr(Hou,Bus)*. We expect these variables to have negative coefficients because the positive correlation between two background risks (e.g., labor and housing) exacerbates the overall risk exposure and hence reduces a household's willingness to bear stock risk.

Overall, our background risk measures consist of 12 variables. We consider a linear regression of stock investment on these variables. We further include three dummy variables, denoted by D_Lab , D_Hou , and D_Bus that respectively indicate if a household has labor income, housing, and business income in a given year to capture the change of background risks over the life cycle. The empirical model is specified as follows:

$$\begin{aligned} StockInvt_{i,t} &= a_0 + a_1Std(Lab_{it}) + a_2Corr(R_{st}, Lab_{it}) + a_3Corr(R_f, Lab_{it}) \\ &+ a_4Std(Hou_{it}) + a_5Corr(R_{st}, Hou_{it}) + a_6Corr(R_{ft}, Hou_{it}) \\ &+ a_7Std(Bus_{it}) + a_8Corr(R_{st}, Bus_{it}) + a_9Corr(R_f, Bus_{it}) \\ &+ a_{10}Corr(Lab_{it}, Hou_{it}) + a_{11}Corr(Lab_{it}, Bus_{it}) + a_{12}Corr(Bus_{it}, Hou_{it}) \\ &+ a_{13}D_{Lab_{it}} + a_{14}D_{-Hou_{it}} + a_{15}D_{-Bus_{it}} + \deltaControls_{it} + \varepsilon_{it}. \end{aligned}$$
(1)

where subscripts *i*, and *t* denote household and year, respectively; *StockInvt* is either a binary variable of stock market participation or a ratio of stock to wealth; *Lab*, *Hou*, *Bus* are growth rates of labor income, home equity and business income, respectively; R_{c} and R_{f} are gross return

rates of a stock market portfolio and the risk-free asset, respectively; and *Controls* is a vector of control variables.

1.2 Control Variables

We follow the prior literature to add control variables. Numerous papers document that race, income, wealth and education each has a positive impact on stock market participation (e.g., Mankiw and Zeldes, 1991; Vissing-Jorgensen, 2002; Hong *et al.*, 2004; and Campbell, 2006). The level of education is regarded as a proxy for fixed entry and transaction costs and is found to be positively significantly related to stock market participation in previous studies. We use two dummy variables, *HSchool* (equal to 1 if the head of household has a high school education) and *College* (equal to 1 if the head of household has a college education), to control for education effects. We expect Log(Age), the log transformation of the age of the household head, to have a positive sign and $(Log(Age))^2$ to have a negative sign to capture the hump-shaped life-cycle pattern of stock holdings (Jagannathan and Kocherlakota, 1996).

Flavin and Yamashita (2002) suggest that the house to net wealth ratio influences a homeowner's portfolio composition. We hence include Log(HsValue) - the log transformation of market value of owner occupied house. Cocco (2005) argues that although housing investment substitutes for stock investment, a mortgage loan serves as a leverage borrowing channel to finance investment in stocks. We include Log(Mortgage) - the log transformation of unpaid mortgage balance as a control variable. Vissing-Jorgensen (2002) documents that the level of nonfinancial income is positively related to stock market participation. We use Log(LabIncome) - the log transformation of labor income as a control variable. To capture the dynamics of labor income risk, we control for unemployment shock by adding a dummy variable *Unemployment*

which equals 1 if the head of household lost its job in a given year and 0 otherwise. We further add a dummy variable that equals 1 if husband and wife work in the same industry and 0 otherwise.

1.3 Econometric Issues and Measurement Errors

We run regressions that relate stock market participation (*DumStk*) and the proportion of stock relative to wealth (*PtfStk*) to a set of explanatory variables. Since *DumStk* is a discrete-choice variable which equals 1 if a household participates in the stock market, and 0 otherwise, we employ the Logit model specified below:

$$Prob(DumStk = 1) = F(\beta'X)$$

$$Prob(DumStk = 0) = 1 - F(\beta'X)$$
where $F(\beta'X) = \frac{e^{\beta'X}}{1 + e^{\beta'X}}$
(2)

Given that a large fraction of households hold no stocks, ordinary least squares regression is not suitable to study the proportion of stock holdings. Several theoretical papers (e.g., Orosel, 1998; Haliassos and Michaelides, 2003; Guo, 2004; Gomes and Michaelides 2005; and Ball, 2008) have treated stock market non-participation (i.e., zero stock holding) as part of a household's portfolio choice. In this framework, agents maximize their life-time utility subject to a budget constraint which includes a participation cost. Consistent with this line of reasoning and following the empirical methodology employed by Guiso *et al.* (1996), Hochoguertel (2002) and Cocco (2005), we adopt a Tobit model where the lower limit is 0 (a household holds no stock).⁶ The Tobit model is specified as follows:

$$PtfStk = \begin{cases} \beta' X + \varepsilon, & \text{if } PtfStk > 0\\ 0, & \text{if otherwise} \end{cases}$$
(3)

An alternative method of estimating the determinants of stock holdings is the Heckman selection model. The selection model suggests that households first make a decision on whether to participate in the stock market; then, conditional on participation, choose the optimal stock holdings related to wealth.⁷ Vissing-Jorgensen (2002) employs a Heckman model with a fixed participation cost. Therefore, for a robustness check, we consider the Heckman model, as specified below:

Participation Equation:
$$S^* = \beta' X + v$$
, $DumStk = \begin{cases} 1, & \text{if } S^* > 0 \\ 0, & \text{otherwise} \end{cases}$ (4)
Stockholding Equation: $PtfStk = \beta' X + \varepsilon$, observed if $DumStk = 1$

where S^* is a latent variable about stock market participation. We only observe a binary variable of *DumStk* and *PtfStk* when a household chooses to participation in stock market. In estimation, we include the lagged stock participation decision and stock holdings as additional explanatory variables in the participation and stockholding equations, respectively.

We further consider an OLS regression using a truncated sample consisting of only stockholders. We include households which hold stock in the current year or ever held stocks in previous years. We then run the OLS regression for this subsample to shed light on how background risks affect stock holdings conditional on participation.

Given a large number of households and a limited number of years in our data, it is hard to estimate the panel regression with household-specific fixed effects. We therefore use year dummy variables to control for time effect, and estimate the standard errors with clustering by individuals in order to correct for serial correlations (a household that holds stocks in the previous year is more likely to hold stocks in the current year).⁸ Massa and Simonov (2006) argue that households tend to hold stocks that are geographically and professionally close to them. We hence control for industry and state fixed effects. The industry dummies are based on the major industry in which the head of household works.⁹

In our baseline model, we assume that background risks are *time-invariant*. In reality, a household's background risks may change over the life cycle. As a robustness check, we consider *time-varying* background risk factors using rolling-over windows (see estimation details in the data section). Using a rolling-over window increases measurement errors because it uses a shorter period rather than the full sample to estimate the standard deviations and correlations.

We further consider a cross-sectional regression. In particular, we regress the timeaverage of stock holdings relative to wealth on the time-invariant background risk factors and the time-averages of other control variables. As for the stock market participation, we sum the stock participation dummy variable over years and divide this variable by the number of years when the household appears in the sample. We classify a household as a stock market participant if it holds stocks in more than half of the time in the sample period.

We assume that the background risk variables are predetermined because adjustments in labor supply, housing and private business are much harder than adjustments in stock investment. In principle, background risk variables can be endogenously determined (e.g., Bodie *et al.*, 1992; and Roussanov, 2004) by risk attitude and investment into human capital. We do not control for risk attitude in our baseline model.¹⁰ However, we believe that our specification is robust to the existence of endogeneity as endogeneity would bias our results towards not finding the expected relationship between the background risk variables and the investment choice. For example, a more risk-averse household would choose to invest in safer assets and select a safer occupation (with a lower standard deviation of labor income), resulting in a positive relationship between the standard deviation of labor income and stock investment. Since our testing hypothesis

predicts that the standard deviation of labor income is negatively related to stock holdings, our regressions provide a conservative estimate of the impacts of these background risk factors on stock market participation and stock holdings.

2. DATA AND DESCRIPTIVE STATISTICS

Our data are drawn from the Panel Study of Income Dynamics (PSID), which is an annual survey maintained by the University of Michigan. The surveys are conducted every year from 1968 to 1997 and every other year after 1997.¹¹ We utilize the PSID surveys from 1975-2009. The long panel with detailed demographic, income, and housing data allows us to construct various measures of income and housing risks. A limitation of the PSID data is that detailed wealth composition such as stock holdings is provided in the Wealth Supplement Survey which was conducted once every five years from 1984 to 1999 and then every other year after 1999. Therefore, the financial asset holdings information is available for these nine years (1984, 1989, 1994, 1999, 2001, 2003, 2005, 2007, and 2009). Since questions related to income and wealth in the PSID data are retrospective¹² (for instance, those asked in 1994 refer to the 1993 calendar year), we refer our sample years as 1983, 1988, 1993, 1998, 2000, 2002, 2004, 2006, and 2008. We use the CRSP NYSE/AMEX/NASDAQ value-weighted market index return as a proxy for risky stock return, and the 30-day T-bill return as a proxy for the risk-free rate. All monetary variables are in constant 1992 dollars using the Consumer Price Index obtained from CRSP.

2.1 Stock Values and Stock Participation

In PSID, stock market participation (denoted by *DumStk*) and the value of stock holdings are self-reported in the surveys. Unfortunately, PSID changed the definition of stock in 1999. Up to the 1997 survey, reported stock holdings include stocks held directly or held in mutual funds, investment trusts, and pension funds. Since the 1999 survey, the value of stock holdings in pension funds is excluded. This change in definition causes an inconsistency in our stock values and stock participation variables over time. We therefore make the following adjustments using questions asked by PSID about pension accounts. The questions are "Do (you/you or anyone in your family) have any money in private annuities or Individual Retirement Accounts (IRAs)?", "Are they mostly in stocks, mostly in interest earning assets, split between the two, or what?", and "How much would they be worth?" We assume that all investments in IRAs are stocks if most money in IRAs is invested in stocks. If a household reports that the money in IRAs is split between stocks and interest sensitive assets, we assume that half of the value in the IRAs is in stocks and the other half is in savings. We then adjust the post-1999 stock variable by summing the reported stock value and the estimated stock value in pension funds.¹³

Previous studies suggest that the properties of portfolio composition relative to demographic variables are sensitive to the way wealth is measured (see, e.g., Heaton and Lucas, 2000a). In computing the proportion of stock value relative to wealth, we consider three definitions of wealth: (i) total family financial wealth—the sum of stock, savings and bond values; (ii) total family wealth without home equity—the sum of values of financial assets, business, vehicles and real estate excluding owner-occupied house minus total debts owed; and (iii) total family wealth with home equity—the sum of value of financial assets, business, vehicles and real estates including home equity of owner-occupied house minus total debts owed. Home equity is the net worth of self-reported market value of house minus unpaid mortgage

balance. These three measures are denoted as $PflStk_1$, $PflStk_2$ and $PflStk_3$, respectively. Our main measure is the stockholding relative to financial wealth ($PflStk_1$). In robustness checks, we also consider $PflStk_2$ and $PflStk_3$ and obtain similar results.

2.2 Background Risk Measures

2.2.1 Time-invariant Background Risk Measures

To create individual background risk measures, we use the 1976-2009 PSID Family Income Files. We generate the 21-year consecutive time series (1976-1997) of annual growth rates of labor income, housing value and business income. Unlike the financial assets which are reported only in the PSID Wealth Supplement Survey, the market value of house and unpaid mortgage are provided in the PSID Family Income Survey. For the period 1997-2009, PSID provides data every other year. We estimate the annual growth rates by dividing the two-year growth rates by two. Overall, we obtain annual growth rates of income and housing value for 27 years.¹⁴ Since PSID does not provide total family business income before 1993, we use the head of household business income as a proxy for total family business income. To make the labor income and business income measures comparable, we also use the head of household labor income as a proxy for total family labor income. We define the head of household business income as the sum of business income from assets and business income from labor.¹⁵ We use home equity - the difference between self-reported house value and unpaid mortgage balance - as our proxy for housing value, because home equity truly reflects a household's wealth accumulation through housing investment. We also use the growth rate of self-reported market value of owner-occupied house (i.e., ignoring unpaid mortgage balance) to redo our regressions.¹⁶ Using the annual growth rates, we calculate for each household the standard

deviations of labor income, home equity and business income, i.e., Std(Lab), Std(Hou) and Std(Bus), and the correlations of these growth rates with stock returns, $Corr(R_{s},.)$, and with the risk-free rate, $Corr(R_{f},.)$. We also calculate the correlations among the three growth rates, Corr(Lab,Hou), Corr(Lab,Bus) and Corr(Bus,Hou).

To minimize measurement errors in the data, we apply several filters to the growth rates of labor income, home equity, and business income. Our baseline analysis requires a household to have at least three years of gross growth rates ranging between 0.5 and 2 to calculate the standard deviation and correlation statistics.¹⁷ That is, we ignore those observations with incomes dropping more than a half or more than doubling in a year because these figures seem implausible and are more likely subject to coding or other errors. This filter is denoted by *Filter2*. To check for robustness, we also require the gross growth rates to lie within the 0.3-3, and 0.2-5 ranges, and denote these filters by *Filter3*, and *Filter5*, respectively.

Note that these filters do not delete households with extraordinary changes of background risks, but they only require a household to have reasonable annual growth rates for at least three years to be included in our sample. For example, a household may continuously provide labor income but suddenly reports a 0 labor income in year t due to unemployment. This will yield a 0 gross growth rate of labor income in year t, and an infinite growth rate in year t+1. We will exclude the observations in years t and t+1 and use the observations for the rest of the years to estimate its labor income risk. We then set the labor income risk variable to 0 for years t and t+1. We use a dummy variable "if head has a job" to control for households without labor income risk. These households could be students, retirees, or self-employed people. We use a dummy variable "if head is unemployed" to control for unemployment shock. For households which do not have

a house or private business, we set the housing risk or business risk to 0. We include dummy variables "if owns a house" and "if owns a business" in our regressions.

2.2.2 Rolling-over Background Risk Measures

The above method to calculate standard deviations and correlations assumes that background risks are time-invariant. In principle, these risks can fluctuate with general economic conditions and can change over the life cycle of a household. Our measures introduced above only capture the variation of background risks across households, but not the time variation of background risks for a given household over its life cycle. To capture the time-series variation of background risks, we employ two rolling-over methods.

First, we consider a household which makes its portfolio choice based on its current and past experience of income and housing value fluctuations, so we employ a backward rolling-over measure. These measures are calculated using prior eight-year data. For example, risk measures in 1983 are calculated using data from 1976 to 1983, and those in 1997 are calculated using data from 1990 to 1997.

Second, rational expectations theory suggests that a household should make its portfolio choice based on its *ex ante* expectation of background risks. We therefore estimate forward rolling-over measures using five-year posterior data. For example, forward risk measures in 1983 are calculated using data from 1983 to 1987, and those in 1993 are calculated using data from 1993 to 1997. The shortening in the number of years used in calculation increases estimation errors. Thus, our main results are based on the time-invariant measures and we provide a robustness check based on the two rolling-over measures.

2.3 Descriptive Statistics

Combining the estimated background risk measures with stock holdings data from the PSID Wealth Supplement Survey, we construct a 9-year unbalanced panel of 4,756 households with 22,610 year-household observations. We confirm the well-known fact of limited stock market participation. The participation rates have significantly increased over the past two decades from 28 percent in 1983 to 44 percent in 2008. However, more than half of the U.S. households still do not hold any stocks. The average ratio of stocks to financial wealth generally increases from 43 percent in 1983 to 51 percent in 2008, with two dips during 2000-2002 and in 2008 which reflect respectively the internet bubble crash and the subprime mortgage crisis.¹⁸

In Figure 1, we present the cross-sectional variation of our 12 background risk variables. In each panel, we display the estimated density using the sample of households who are exposed to the corresponding risks and the normal density curve for ease of comparison. There is a large dispersion for each of these risk factors. The distribution of the standard deviation of labor income growth is skewed to the right, indicating that most households are exposed to moderate labor income risk while a small fraction is subject to large labor risk. A similar pattern is observed for the standard deviation of housing growth. We see a spike around zero in the empirical density for the correlations of income growth with stock returns and with the risk-free rate, indicating that there are a substantial number of households whose income growth is unrelated with stocks returns and with the risk-free rate. We find a similar pattern for the correlations of housing growth with financial asset returns. The distributions of both correlations of housing income with labor income and with business income are well dispersed with a spike in zero. On the other hand, for most households, the correlation between labor and business incomes is close to zero. This result is expected as households owning a private business are mostly self-employed and have no labor income.

Panel A of Table 2 reports the summary statistics of the variables used in our regressions. The top part of this table summarizes stock investment. For example, overall 38.8 percent of households participate in stock market. The middle part in Panel A presents the summary statistics of the background risk variables. We observe substantial *heterogeneity* of background risks across households. Overall, 83.6 percent of households have labor income, 78.7 percent of households own a house, but only 16 percent have a private business.¹⁹ Within the group of households that have labor income, the median standard deviation of labor income growth is 19.5 percent. Similarly, for households with business income, the median standard deviation of business income growth is 21.6 percent. The bottom part of Panel A reports some demographic information. The average age of the heads of households is 48.196; the average family size is 2.682; the average family income is \$52,644; 30.9 percent of the heads of households have a college degree, while 52.7 percent have only a high school education.

Panel B of Table 2 presents the correlation matrix of the 12 background risk measures. The correlation between a background risk measure and stock returns is closely related to its correlation with the risk-free rate, suggesting some degree of multicollinearity. We therefore adjust our baseline model by excluding the correlations between the risk-free rate and background risk assets and obtain similar results as our baseline analysis. The results using these measures are not reported but are available upon request.²⁰

3. EMPIRICAL RESULTS

3.1 Statistical Significance

Table 3 presents maximum likelihood estimates of the Logit regressions. Five specifications are estimated, each with a different combination of the three types of background risks: (1) no background risk; (2) with only labor income risk; (3) with both labor income and housing risks; (4) with both labor income and business risks; and (5) with all three types of background risks. The top panel reports log likelihood ratio tests for various model comparisons.

Column (1) displays our benchmark model without considering background risks. We find strong explanatory power of education, race, income and wealth for stock market participation, confirming the results of earlier studies. The positive coefficient on Log(Age) and the negative coefficient on $(Log(Age))^2$ confirm the hump-shape pattern of stock market participation with age although these variables are not statistically significant. We find that the "house value" variable carries the expected negative sign and "unpaid mortgage" has the expected positive coefficient, consistent with Cocco (2005) and Campbell (2006). These parameters are statistically significant at the 1 percent level. These results suggest that although housing investment crowds out stock investment, mortgage loans can be used as a financing channel to support stock investment. The "head has a job" variable is positively related to stock market participation, but the effect is not statistically significant. The "owns a house" variable significantly increases the likelihood of stock market participation whereas the "has a business" variable significantly reduces participation. The "head in unemployment" is negatively related to participation but it is not statistically significant. "Head and wife in same industry" significantly reduces participation.

In Column (2), we add the three labor income risk variables to the benchmark model. The coefficients of these three variables are estimated with the expected signs and are statistically significant at the 1 percent level. They imply that a household is more (less) likely to enter the

stock market if its labor income is less (more) uncertain, if its labor income is less (more) highly correlated with stock return, or if its labor income is more (less) highly correlated with the riskfree rate. Both $Corr(R_s, Lab)$ and $Corr(R_f, Lab)$ are statistically significant but with opposite effects on stock investment, suggesting that labor income risk can affect a household's stock investment in different ways. This result is consistent with various simulation studies which document that labor income reduces stock holdings when it is modeled as a risky asset, whereas it encourages stock investment when it is regarded as a risk-free asset. We conduct a log likelihood ratio test to investigate whether specification (2) outperforms specification (1). Given the chi-square statistic of 32 with degrees of freedom of 3, we reject specification (1) in favor of specification (2) at the 1 percent significance level.

Column (3) studies housing risk after controlling for labor income risk. All four parameters associated with housing risk are estimated with the expected signs. The coefficient of $Corr(R_s,Hou)$ is not statistically significant, while the other three variables associated with housing risk are significant at 10 percent level or higher. Std(Hou) is significantly negatively related to stock market participation at the 1 percent level with the coefficient -0.453. The variable $Corr(R_f,Hou)$ is positively related to stock market participation. This finding is consistent with the prior literature suggesting that real estate investment is a good hedge against inflation (e.g., Goetzmann and Valaitis, 2006). It is interesting to note that the correlation between labor income and home equity Corr(Lab,Hou) carries a significantly negative sign. This result suggests that the comovement of housing and labor income increases risk exposures, thus reducing the household's willingness to participate in the stock market. Our log likelihood ratio test rejects specification (2) in favor of specification (3) at the 1 percent significance level, suggesting the importance of housing risk in stock market participation decision.

Column (4) shows that the standard deviation of business income has a negative impact on stock participation at the 10 significance level. Both the correlations of business income with the risk-free rate and with stock returns are insignificant. The log likelihood ratio test does not reject specification (2) in favor of (4) given the chi-square statistic of 5, suggesting that the overall impact of business risk on household stock market participation is not statistically significant.

In Column (5), we report the results when all three types of background risks are jointly considered. All background risk variables that are significant in previous regressions continue to be statistically significant. Furthermore, this model outperforms specification (1) at the 1 percent significance level. Based on the likelihood ratio test, the three types of background risks are important to a household's decision to participate in the stock market.

Table 4 studies the potential of background risks to explain the heterogeneity in portfolio compositions among households using Tobit regressions. The dependent variable, *PflStk_1*, is the ratio of stock to financial wealth. Compared with the results from the Logit regressions in explaining market participation, we find that the variables that are used to capture the background risks continue to have the expected signs in explaining portfolio choice in the Tobit regressions. The likelihood ratio tests confirm our previous findings that labor and housing risks appear to be more important than business risk.

In terms of the relative importance, we find labor income risk to be the most important, followed by housing risk, while business risk is less important. Three labor income risk factors are all statistically significant. The standard deviation of home equity growth, and the correlation between the risk-free rate and home equity growth are significant but the correlation between stock return and home equity growth is not significant. As for the business risk, only the standard

deviation of business income growth is significantly negatively related to stock investment. In addition, the correlation between labor income and home equity is negatively related to stock holdings.

3.2 Economic Significance

Given the statistical significance of the background risk factors presented above, we further study the quantitative impact of these risk factors. For each type of risk, we estimate the change of a household's probability of stock market participation and proportion of stock to financial wealth by assuming that the corresponding risk variables change one standard deviation in the unfavorable direction from their sample means while holding all other variables at their sample means. Table 5 reports the results.

The left part of Panel A reports the change in the probability of stock market participation. The calculation is based on the estimated coefficients in Column (5) of Table 3. For labor income risk, if *Std*(*Lab*), *Corr*(R_s ,*Lab*) and *Corr*(R_f ,*Lab*) all shift one standard deviation from their respective sample means, the household will reduce its likelihood to participate in the stock market by 6.26 percent. Similarly, for housing risk and business risk, the respective changes in probabilities are 3.41 percent and 1.63 percent. If all background risk variables change together, the probability of participation declines by 10.82 percent.²¹

The right part of Panel A of Table 5 provides the economic significance of background risk variables on the proportion of stock holdings. Using the estimated coefficients reported in Column (5) of Table 4, we calculate the change of the proportion of stock holdings relative to financial wealth. For labor income risk, if Std(Lab), $Corr(R_s,Lab)$ and $Corr(R_f,Lab)$ all shift one standard deviation from their respective sample means, the household will reduce its proportion of stock holdings by 1.92 percent. Similarly, for housing risk and business risk, the respective changes are 1.16 percent and 0.47 percent. If all background risk variables change together, the proportion of stock holdings declines by 3.69 percent.

Considering that the average stock market participation rate is 38.8 percent, the 10.82 percent decrease in the probability of participation due to the one standard deviation increase in the three background risks is a 28 percent (=10.82/38.8) reduction in stock market participation in the sample. Similarly, as the sample average ratio of stock to financial wealth is 20.7 percent, the 3.69 percent decrease in stock holdings due to the one standard deviation increase in background risks implies a 18 percent (=3.69/20.7) decline in stock holdings. These figures show that the effect of background risks on stock investments is economically important. The relative impact on market participation is especially pronounced.

In Panel B of Table 5, we examine the impact of education on the relationship between a household's stock market participation and stock holdings with background risks. Consistent with the transaction costs argument, a more highly educated household is more sensitive to a change in its background risks. When the overall background risk increases by one standard deviation, a household without a high school education will decrease its stock market participation probability by 8.43 percent and its proportion of stock holdings by 3.07 percent. In contrast, a household with a college education will reduce its stock market participation probability by 12.53 percent and its proportion of stock holdings by 4.24 percent.

The above results are consistent with the notion that education level is a proxy for transaction costs (fixed entry and information costs) in previous studies (e.g., Campbell, 2006; and Vissing-Jorgensen, 2002). A more highly educated household is more likely to adjust its

stock investment in response to a change in its background risks because its entry and information costs are lower.

3.3 Alternative Measures of Background Risks

Table 6 conducts more tests using alternative measures of background risks. In Columns (1) and (2), we redo our tests using backward rolling-over measures for the Logit regression of stock market participation and for the Tobit regression of stock holdings, respectively. The standard deviation of labor income growth rate reduces stock market participation and the proportion of stock to financial wealth. The correlation between stock returns and labor income growth rate is negatively related to stock market participation and stock holdings whereas the correlation between the risk-free rate and labor income growth is not significant. Housing risk is also important in that the standard deviation of home equity growth significantly decreases stock market participation and stock holdings. The correlation between stock returns and business income growth carries an unexpected positive sign which is statistically significant at the 10 percent level in both regressions.

In Columns (3) and (4), we redo our regressions using forward rolling-over measures. For the stock market participation regression, the only two significant variables are the standard deviation of labor income growth and the correlation between the risk-free rate and labor income growth. As for the stock holdings regression, we find that three variables are statistically significant: the standard deviation of labor income growth, standard deviation of business income growth, and the correlation between stock returns and business income growth.

In Columns (5) and (6), we consider cross-sectional regressions. For the Logit regression in Column (5), we sum the stock market participation dummy variable over years and divide this variable by the number of years when the household appears in the sample. We classify a household as a stock market participant if it holds stocks in more than half of the time in the sample period. As for the Tobit regression in Column (6), we regress the time-average of stock holdings relative to wealth on the time-invariant background risk factors and the time-averages of other control variables.

As shown in Column (5), the standard deviation of labor income growth significantly negatively impacts stock market participation, while the correlation of labor income and the risk-free rate significantly encourages stock market participation. The standard deviation of home equity growth rate also significantly decreases stock market participation. Column (6) confirms the importance of labor income risk variables in affecting a household's portfolio choice. Furthermore, the correlation of home equity growth and the risk-free rate positively impacts investment in stock while the correlation of labor income and housing significantly decreases stock holdings.

In general, these tests using alterative measures of background risks yield weaker results than the baseline cases reported in Tables 3 and 4, primarily due to the less precise estimates of background risk measures with fewer observations. However, some observations can be made. Labor income risk is the most important one. The standard deviation of labor income growth is significant in all regressions. The correlation between stock returns and labor income growth carries the expected negative sign and is significant in 3 out of 6 regressions. The correlation between labor income growth and the risk-free rate also has the expected positive sign in all regressions and is significant in 3 out of 6 regressions. We therefore conclude that the volatility of labor income is an important factor that significantly affects household stock market participation and stock holdings. Moreover, households with risky (risk-free) labor income are

less (more) likely to participation in stock market and to hold less (more) risky assets in their portfolios.

The standard deviation of home equity growth rate carries the expected negative sign and is significant in 3 out of 6 regressions, suggesting that the volatility of home equity growth discourages stock market participation and stock holdings. However, results based on the correlation between home equity growth and financial asset returns are in general insignificant. Furthermore, we find that the business income risk factors which are estimated with the least observations yield the most mixed results.

3.4 Alternative Estimation Methods of Portfolio Choice

In Table 7, we compare the estimation results using the Tobit model, the Heckman selection model and OLS regressions for the truncated sample consisting of only stockholders. We use two additional measures of the stock to wealth ratio: the ratio of stock to total wealth without home equity, denoted as $PflStk_2$, and the ratio of stock to total wealth including home equity, denoted as $PflStk_3$.

Column (1) in Table 8 is a repeat of the Tobit model in column (5) of Table 5 where the stock to wealth ratio is defined as a ratio of stock to total financial wealth (*Pflstk_1*). Columns (2) and (3) in Table 7 are Tobit models using alternative measures of the stock to wealth ratio (*Pflstk_2* and *Pflstick_3*). The results are very similar to previous findings in Table 5.

The next three columns in Table 7 present OLS regressions. In the OLS regressions, Std(Lab) is not significant. As for the correlation terms, $Corr(R_s,Lab)$ is negatively related to stock holdings, and $Corr(R_f,Lab)$ is positively related to stock holdings, consistent with findings in the Tobit regressions. Among the three housing risk variables, only $Corr(R_s,Hou)$ is

significantly negatively related to *PtfStk_1* and *PtfStk_2*. As for the business income risk, *Std(Bus)* has the expected negative sign, but it is statistically significant only in the *PtfStk_3* regression. Overall, the OLS regression results are qualitatively consistent with those using the Tobit model, but are statistically less significant.

As for Heckman selection model, we find that lagged stockholding has strong explanatory power for current stock holding. We report the "*Lambda*", which would be zero if stockholders were a random subgroup of the population. We reject the hypothesis that stockholders are random subgroup in favor of the alternative that stockholders are a selected group. Labor income risk appears less important in the Heckman model. The only significant variable is Std(Lab), which is at the 10% significance for $PtfStk_3$. In contrast, housing risk is relatively more important after controlling for sample selection. In the $PtfStk_3$ regression, Std(Hou) carries an unexpected positive sign which is significant at the 10% level; $Corr(R_s, Hou)$ is negatively related to stock holdings; and $Corr(R_f, Hou)$ is positively related to stock holdings. As for business risk, consistent with previous findings, only Std(Bus) is important. In summary, the estimation of the Heckman model is in general consistent with the Tobit model, but is statistically less significant.

Overall, while we find that the results using OLS and Heckman models are weaker than those using the Tobit model, they still indicate the importance of background risks. Therefore, the significance of background risks in portfolio holdings is not primarily driven by the difference between stockholders and non-stockholders. We also note that the impact of background risks is relatively more important for stock participation than for stock holdings.

3.5 Other Robustness Checks

We conduct additional robustness checks. We first consider alternative ways to construct background risk factors. Specifically, we examine separate effects based on whether the correlation of a background risk factor with stock returns is positive or negative, i.e., the explanatory variables are (standard deviation × positive correlation) and (standard deviation × negative correlation). This specification allows us to further examine the hedging motive hypothesis. We find that the standard deviation of labor income interacting with a positive correlation, $(Std(Lab) \times Corr(R_s, Lab)^+)$, is significantly negatively related to stock participation while the impact of the standard deviation of labor income interacting with a negative correlation, $(Std(Lab) \times Corr(R_s, Lab)^-)$, is not significant. We also consider the correlations of excess stock returns with the background risk variables, i.e., $(Corr(R_s-R_f, X))$, where X is labor income, home equity, or business income growth. The results using this alternative measures are consistent with our baseline regressions.

It is known that estimation of Tobit models can be sensitive to the underlying assumptions about the error terms and indeed maximum likelihood estimation can be inconsistent under heteroscedasticity or nonnormality (Amemiya, 1985, pp.378-381). We adopt three alternative specifications, which assume the residual standard errors to be an exponential function of total wealth, or total income, or both, respectively. These experiments produce similar results.

Since the PSID changes the definition of stock holdings in 1997. As a robustness check, we redo our baseline model by excluding observations for the years 1997 and 1999 and the qualitative results stay the same. Our baseline study uses *filter2* to estimate background risk variables. As a robustness test, we apply *filter3* and *filter5* to filter our data and obtain similar results. We also redo our regressions using three sub-samples of households that have labor

income, housing, and business income, respectively. The results in these experiments confirm our previous findings. Following Angerer and Lam (2009), we decompose the background risks to expected (predictable) and unexpected (unpredictable) components. Specifically, we regress the growth rate of labor income (home equity, business income) on the lagged dependent variable and other explanatory variables, including the age of the head of household and its squared term, the age of wife and its squared term, and the two education variables. The regression residuals are regarded as the unexpected component and the fitted values of the regression are considered as the expected component of the background risk. We find that the expected labor income risk is positively related to stock market participation and stock holdings while the unexpected labor income risk is negatively related to stock market participation and stock holdings, consistent with Angerer and Lam (2009). We obtain a similar result for housing income risk. In contrast, the expected business income risk decreases stock investment while the unexpected business risk increases stock investment.

Results obtained in this section are not reported to save space but are available upon request.

4. CONCLUSIONS

Using a sample of U.S. households with individual background risk measures, we examine the empirical importance of background risks for a household's investment decision. We document significant heterogeneity of background risk exposures across households. The low stock market participation rates and the large variation of stock holdings are significantly related to the heterogeneity in background risks across households. Specifically, a household is more (less) likely to enter the stock market and invests a larger (smaller) fraction of wealth in

stocks if its non-financial income (e.g., labor income) is less (more) volatile, is less (more) highly correlated with stock returns, or is more (less) highly correlated with the risk-free rate. In terms of relative importance, we find that labor income risk is the most important, followed by housing risk, while the impact of business income risk is limited.

References

Amemiya, Takeshi. (1985) Advanced Econometrics, Cambridge: Harvard University Press.

- Angerer, Xiaohong, and Pok-Sang Lam. (2009) "Income Risk and Portfolio Choice: An Empirical Study." *Journal of Finance*, 64, 1037-1055.
- Ball, Steffan. (2008) "Stock Market Participation, Portfolio Choice and Pension over the Life Cycle." *mimeo, Finance and Economics Discussion Series* 2008-64, Washington: Board of Governors of the Federal Reserve System.
- Benzoni, Luca, Pierre Collin-Dufresne, and Robert S. Goldstein. (2007) "Portfolio Choice over the Life-Cycle When the Stock and Labor Markets Are Cointegrated." *Journal of Finance*, 62, 2123-2167.
- Bodie, Zvi, Robert Merton, and William F. Samuelson. (1992) "Labor Supply Flexibility and Portfolio Choice in a Life Cycle Model." *Journal of Economic Dynamics and Control*, 16, 427-449.
- Campbell, John Y. (2006) "Household Finance." Journal of Finance, 61, 1553-1604.
- Chen, Xuanjuan, Tong Yao, and Tong Yu. (2007) "How Does Background Risks Affect Investment Risk-Taking? Evidence from Insurers' Corporate Bond Portfolios." *mimeo*, University of Iowa.
- Cocco, Joao F. (2005) "Portfolio Choice in the Presence of Housing." *Review of Financial Studies*, 18, 535-567.
- Cocco Joao F., Francisco J. Gomes, and Pascal J. Maenhout. (2005) "Consumption and Portfolio Choice over the Life Cycle." *Review of Financial Studies*, 18, 491-532.
- Cochrane, John. (2006) "Financial Markets and the Real Economy." in *International Library of Critical Writings in Financial Economics*, Vol 18, Ch 7, ed. by J. Cochrane, London: Edward Elgar.
- Cochrane, John. (2008) "A Mean-Variance Benchmark for Intertemporal Portfolio Choice." *mimeo*, University of Chicago.
- Constantinides, George M., and Darrell Duffie. (1996) "Asset Pricing with Heterogeneous Consumers." Journal of Political Economy, 104, 219-240.
- Davis, Steven J., and Paul Willen. (2002) "Risky Labor Income and Portfolio Choice." in Zvi Bodie, Olivia S. Mitchell, Brett Hammond, and Stephen Zeldes, eds., *Innovations in Financing Retirement*, University of Pennsylvania Press.
- Dimmock, Stephen G. (2012) "Background Risk and University Endowment Funds." *Review of Economics and Statistics*, 94, 789-799.
- Duffie, Darrell, W. Fleming, HM Soner, and Thaleia Zariphopoulou. (1997) "Hedging in Incomplete Markets with HARA Utility." *Journal of Economic Dynamics and Control*, 21, 753-782.
- Eiling, Esther. (2013) "Industry Specific Human Capital, Idiosyncratic Risk and the Cross-Section of Expected Stock Returns." *Journal of Finance*, 68, 43-84.
- Flavin, Marjorie, and Takashi Yamashita. (2002) "Owner-Occupied Housing and Composition of the Household Portfolio." *American Economic Review*, 92, 345-362.

- Goetzmann, William N., and Eduardas Valaitis. (2006) "Simulating Real Estate in the Investment Portfolio: Model Uncertainty and Inflation Hedging." mimeo, Yale University.
- Gomes, Fransico, and Allexander Michaelides. (2005) "Optimal Life-Cycle Asset Allocation: Understanding the Empirical Evidence." *Journal of Finance*, 55, 869-904.
- Guiso, Luigi, Tullio Jappell, and Daniele Terlizzese. (1996) "Income Risk, Borrowing Constraints, and Portfolio Choice." *American Economic Review*, 86, 158-172.
- Guo, Hui. (2004) "Limited Stock Market Participation and Asset Prices in a Dynamic Economy." Journal of Financial and Quantitative Analysis, 39, 495-516.
- Haliassos, Michael, and Carol C. Bertaut. (1995) "Why Do So Few Hold Stocks?" *Economic Journal*, 105, 1110-1129.
- Haliassos, Michael, and Alexander Michaelides. (2003) "Portfolio Choice and Liquidity Constraints." International Economic Review, 44, 143-177.
- Heaton, John, and Deborah Lucas. (1996) "Evaluating the Effects of Incomplete Markets on Risk Sharing and Asset Pricing." *Journal of Political Economy*, 104, 443-487.
- Heaton, John, and Deborah Lucas. (1997) "Market Frictions, Savings Behavior, and Portfolio Choice." *Macroeconomic Dynamics*, 1, 76-101.
- Heaton, John, and Deborah Lucas. (2000a) "Portfolio Choice in the Presence of Background Risk." *Economic Journal*, 110, 1-26.
- Heaton, John, and Deborah Lucas. (2000b) "Portfolio Choice and Asset Prices: The Importance of Entrepreneurial Risk." *Journal of Finance*, 55, 1163-1198.
- Hochguertel, Stefan. (2002) "Precautionary Motives and Portfolio Decisions." Journal of Applied Econometrics, 18, 61-77.
- Hong, Harrison, Jeffery D. Kubik, and Jeremy C. Stein. (2004) "Social Interaction and Stock-Market Participation." *Journal of Finance*, 59, 137-163.
- Jagannathan, Ravi, and Narayana Kocherlakota. (1996) "Why Should Older People Invest Less in Stocks than Younger People?" *Federal Reserve Bank of Minneapolis Quarterly Review*, 20, 11-23.
- Jagannathan, Ravi, and Zhenyu Wang. (1996) "The Conditional CAPM and the Cross-Section of Expected Returns." *Journal of Finance*, 51, 3-53.
- Krueger, Dirk, and Hanno Lustig. (2010) "When Is Market Incompleteness Irrelevant for the Price of Aggregate Risk (and When Is It Not)?" *Journal of Economic Theory*, 145, 1-41.
- Mankiw, N. Gregory, and Stephen P. Zeldes. (1991) "The Consumption of Stockholders and Non-Stockholders." *Journal of Financial Economics*, 29, 97-112.
- Massa, Massimo, and Andrei Simonov. (2006) "Hedging, Familiarity and Portfolio Choice." *Review of Financial Studies*, 19, 633-685.
- Orosel, Gerhard O. (1998) "Participation Costs, Trend Chasing, and Volatility of Stock Prices." *Review of Financial Studies*, 11, 521-557.
- Petersen, Mitchell A. (2009) "Estimating Standard Errors in Finance Panel Data Sets: Comparing Approaches." *Review of Financial Studies*, 22, 435-480.

- Roussanov, Nikolai. (2004) "Human Capital Investment and Portfolio Choice over the Life-Cycle." *mimeo*, University of Chicago.
- Storesletten, Kjetil, Chris I. Telmer, and Amir Yaron. (2004) "Cyclical Dynamics in Idiosyncratic Labor-Market Risk." *Journal of Political Economy*, 112, 695-717.
- Viceira, Luis M. (2001) "Optimal Portfolio Choice for Long-Horizon Investors with Nontradable Labor Income." *Journal of Finance*, 56, 433-470.
- Vissing-Jorgensen, Annette. (2002) "Towards an Explanation of Household Portfolio Choice Heterogeneity: Nonfinancial Income and Participation Cost Structures." *NBER Working Paper*.
- Yao, Rui, and Harold Zhang. (2005) "Optimal Consumption and Portfolio Choice with Risky Housing and Borrowing Constraints." *Review of Financial Studies*, 18, 197-239.

Table 1 Definition and Expected Impact of Background Risk Variables

This table presents the definition of background risk variables and their expected impact on stock market participation and stockholdings.

Variable	Definition	Predicted
		sign
	Labor Income Risk	
D_Lab	Dummy variable equal to 1 if a household has labor income in a given year	+
Std(Lab)	Standard deviation of annual growth rate of labor income	-
$Corr(R_s, Lab)$	Correlation between the annual growth rate of labor income and stock return	-
$Corr(R_f, Lab)$	Correlation between the annual growth rate of labor income and riskfree rate	+
	Housing Risk	
D_Hou	Dummy variable equal to 1 if a household owns house in a given year	+
Std(Hou)	Standard deviation of annual growth rate of home equity	-
$Corr(R_s, Hou)$	Correlation between the annual growth rate of home equity and stock return	-
$Corr(R_f, Hou)$	Correlation between the annual growth rate of home equity and riskfree rate	+
	Business Income Risk	
D_Bus	Dummy variable equal to 1 if a household has business income in a given year	-
Std(Bus)	Standard deviation of annual growth rate of business income	-
$Corr(R_s, Bus)$	Correlation between the annual growth rate of business income and stock return	-
$Corr(R_f, Bus)$	Correlation between the annual growth rate of business income and the riskfree rate	+
	Correlations among Three Types of Background Risks	
Corr(Lab,Hou)	Correlation between the annual growth rates of labor income and home equity	-
Corr(Lab,Bus)	Correlation between the annual growth rates of labor and business income	-
Corr(Hou,Bus)	Correlation between the annual growth rates of home equity and business income	-

Table 2 Characteristics of Explanatory Variables

Panel A reports summary statistics of the explanatory variables in our regressions. Panel B reports the correlation matrix of background risk variables. Std(X) is standard deviation of variable X; Corr(X,Y) is correlation between X and Y; *Lab*, *Hou* and *Bus* are, respectively, annual growth rates of labor income, home equity and business income; R_s is annual gross return of CRSP value-weighted market index; and R_f is annual gross return of the 30-day T-bill. The data are a 9-year unbalanced panel for 4,756 households for the years 1984, 1989, 1994, 1999, 2001, 2003, 2005, 2007, and 2009. "***", "**" and "*" denote statistical significance at the 1, 5 and 10 percent levels, respectively.

	Panel A: Summary Statistics									
	Obs.	Mean	Median	Std. Dev.	Min	Max				
Stock Investment										
Stock participation	22,610	0.388	0	0.487	0	1				
Stock to financial wealth	22,610	0.207	0	0.320	0	1				
Stock to wealth without home equity	22,610	0.145	0	0.261	0	1				
Stock to wealth with home equity	22,610	0.091	0	0.178	0	1				
Background Risks										
If head of household works	22,610	0.836	1	0.370	0	1				
Std(Lab)	18,907	0.206	0.195	0.106	0	0.718				
Corr(R _s , Lab)	18,907	0.030	0.026	0.277	-0.791	0.801				
Corr(R _f , Lab)	18,907	0.046	0.04	0.274	-0.789	0.861				
If household owns a house	22,610	0.787	1	0.410	0	1				
Std(Hou)	17,787	0.337	0.314	0.184	0	1.887				
Corr(R _s , Hou)	17,787	0.029	0.013	0.272	-0.792	0.826				
Corr(R _f , Hou)	17,787	-0.005	0	0.274	-0.812	0.814				
If household owns a business	22,610	0.160	0	0.367	0	1				
Std(Bus)	3,611	0.291	0.216	0.313	0	1.544				
Corr(R _s , Bus)	3,611	0.034	0	0.248	-0.737	0.808				
Corr(R _f , Bus)	3,611	0.038	0	0.264	-0.764	0.787				
Corr(Lab, Hou)	14,090	0.014	0	0.312	-0.783	0.796				
Corr(Lab, Bus)	3,469	-0.004	0	0.183	-0.748	0.748				
Corr(Hou, Bus)	3,121	0.009	0	0.258	-0.808	0.816				
Control Variables										
Head age	22,610	48.196	46	15.691	18	101				
Family size	22,610	2.682	2	1.337	1	12				
Race	22,610	0.919	1	0.273	0	1				
High school education	22,610	0.527	1	0.499	0	1				
College education	22,610	0.309	0	0.462	0	1				
Total wealth including home equity	22,610	173,428	82,674	254,124	-36,302	2,093,793				
Total before-tax family income	22,610	52,644	44,558	37,882	39	258,391				
Total head labor income	22,610	29,785	25,304	29,153	0	253,671				
House value	22,610	107,544	81,552	116,948	0	1,731,544				
Unpaid mortgage value	22,610	33,724	8,896	46,456	0	263,478				
If head and wife in same industry	22,610	0.063	0	0.244	0	1				
If head is in unemployment	22,610	0.061	0	0.24	0	1				

		(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)	(10)	(11)	(12)
Std(Lab)	(1)	1.000	0.034***	0.027***	-0.073***	-0.045***	-0.032****	0.094***	-0.001	0.001	0.014^{**}	-0.025***	0.009
Corr(R _s ,Lab)	(2)		1.000	0.434***	-0.014**	0.020^{***}	0.010	-0.025***	-0.003	-0.003	0.020^{***}	0.014^{**}	-0.006
Corr(R _f ,Lab)	(3)			1.000	0.007	0.014^{**}	-0.002	-0.038***	-0.018***	-0.017**	0.027^{***}	0.015**	0.003
Std(Hou)	(4)				1.000	-0.015**	-0.054***	0.072^{***}	0.017^{**}	0.028^{***}	0.002	0.003	0.021***
Corr(R _s ,Hou)	(5)					1.000	0.492***	-0.009	-0.013*	-0.013*	0.022^{***}	-0.004	-0.008
Corr(R _f ,Hou)	(6)						1.000	-0.009	-0.005	-0.005	0.059***	-0.014**	0.003
Std(Bus)	(7)							1.000	0.167^{***}	0.211***	0.022^{***}	-0.026***	0.021***
Corr(R _s ,Bus)	(8)								1.000	0.435***	0.015^{**}	-0.023***	0.088^{***}
Corr(R _f ,Bus)	(9)									1.000	0.014^{**}	-0.015**	0.010
Corr(Lab,Hou)	(10)										1.000	-0.005	-0.048***
Corr(Lab,Bus)	(11)											1.000	-0.049***
Corr(Hou,Bus)	(12)												1.000

Panel B: Correlation Matrix of Background Risk Factors

Table 3 Determinants of Stock Market Participation

This table reports maximum likelihood estimation of Logit regressions. The dependent variable is *DumStk*, which is a binary-choice variable equal to 1 if a household participates in stock market and 0 otherwise. In each panel, coefficient estimates are reported with associated *t*-statistics in parentheses. Standard errors are estimated with cluster of households. *Log(X)* is natural logarithm of variable X; *Std(X)* is standard deviation of X; *Corr(X,Y)* is correlation between X and Y; *Lab*, *Hou* and *Bus* are, respectively, annual growth rates of labor income, home equity and business income; R_s is annual gross return of CRSP value-weighted market index; and R_f is annual gross return of the 30-day T-bill. "***", "**" and "*" denote statistical significance at the 1, 5 and 10 percent levels, respectively.

			Labor	Labor	
	Baseline	Labor Risk	& House Risk	& Business Risk	All Risks
_	(1)	(2)	(3)	(4)	(5)
Likelihood ratio test		(2)-(1)	(3)-(2)	(4)-(2)	(5)-(1)
Chi-square of likelihood ratio test		32.000	17.000	5.000	55.000
p-value of likelihood ratio test		(0.00)	(0.00)	(0.17)	(0.00)
Std(Lab)		-1.210***	-1.131***	-1.187***	-1.105***
		(-4.36)	(-4.06)	(-4.30)	(-3.99)
Corr(R _s , Lab)		-0.308***	-0.316***	-0.314***	-0.321***
		(-2.91)	(-2.99)	(-2.96)	(-3.04)
Corr(R _f , Lab)		0.283***	0.299***	0.284***	0.298***
		(2.60)	(2.75)	(2.61)	(2.75)
Std(Hou)			-0.453***		-0.458***
			(-2.88)		(-2.91)
Corr(R _s , Hou)			-0.050		-0.047
			(-0.44)		(-0.41)
Corr(R _f , Hou)			0.195*		0.192*
			(1.72)		(1.69)
Std(Bus)				-0.350*	-0.336*
				(-1.86)	(-1.80)
Corr(R _s , Bus)				0.126	0.095
				(0.53)	(0.40)
Corr(R _f , Bus)				0.230	0.258
				(0.92)	(1.04)
Corr(Lab, Hou)			-0.200**		-0.194**
			(-2.18)		(-2.12)
Corr(Lab, Bus)				0.175	0.193
				(0.65)	(0.71)
Corr(Hou, Bus)					0.161
					(0.80)
Head has a job	0.092	0.339**	0.332**	0.370***	0.356***
	(0.75)	(2.48)	(2.43)	(2.71)	(2.60)
Owns a house	3.031***	3.044***	3.330***	3.087***	3.370***
	(6.17)	(6.18)	(6.62)	(6.25)	(6.69)

Has a business	-0.250***	-0.216***	-0.208***	-0.139*	-0.134*
	(-3.69)	(-3.20)	(-3.09)	(-1.73)	(-1.67)
Head in unemployment	-0.109	-0.082	-0.075	-0.082	-0.074
	(-1.29)	(-0.96)	(-0.88)	(-0.95)	(-0.87)
Head and wife in same industry	-0.253***	-0.245***	-0.247***	-0.240***	-0.243***
	(-2.74)	(-2.66)	(-2.70)	(-2.60)	(-2.65)
Log(Age)	1.928	1.609	1.956	1.695	2.052
	(1.14)	(0.95)	(1.15)	(1.00)	(1.20)
$(Log(Age))^2$	-0.308	-0.267	-0.312	-0.278	-0.324
	(-1.36)	(-1.17)	(-1.36)	(-1.22)	(-1.42)
Log(Family size)	-0.219***	-0.218***	-0.212***	-0.222***	-0.216***
	(-4.26)	(-4.22)	(-4.10)	(-4.29)	(-4.15)
Race	0.709***	0.721***	0.723***	0.719***	0.719***
	(7.30)	(7.47)	(7.53)	(7.46)	(7.50)
High school	0.380***	0.370***	0.377***	0.369***	0.374***
	(4.84)	(4.73)	(4.82)	(4.71)	(4.78)
College	0.951***	0.931***	0.932***	0.931***	0.932***
	(10.8)	(10.6)	(10.6)	(10.6)	(10.6)
Log(Income)	0.374***	0.365***	0.369***	0.370***	0.374***
	(9.49)	(9.25)	(9.29)	(9.31)	(9.36)
Log(Wealth)	1.381***	1.387***	1.380***	1.390***	1.383***
	(30.0)	(30.2)	(29.9)	(30.1)	(29.8)
Log(House value)	-0.363***	-0.367***	-0.379***	-0.371***	-0.383***
	(-7.84)	(-7.92)	(-8.15)	(-7.98)	(-8.20)
Log(Unpaid Mortgage)	0.058***	0.057***	0.060***	0.058***	0.060***
	(9.12)	(9.01)	(9.47)	(9.06)	(9.50)
Log(Head Labor Income)	-0.012	-0.011	-0.011	-0.015	-0.014
	(-1.24)	(-1.12)	(-1.14)	(-1.51)	(-1.48)
Constant	Yes	Yes	Yes	Yes	Yes
Industry fixed effect	Yes	Yes	Yes	Yes	Yes
State fixed effect	Yes	Yes	Yes	Yes	Yes
Year fixed effect	Yes	Yes	Yes	Yes	Yes
Observations	21,938	21,938	21,938	21,938	21,938
Pseudo R-square	0.266	0.268	0.269	0.269	0.270

Table 4Determinants of Stock Holdings

This table reports maximum likelihood estimation of Tobit regressions. The dependent variable is $PflStk_1$, which is the proportion of stock relative to total financial wealth. In each panel, coefficient estimates are reported with associated *t*-statistics in parentheses. Standard errors are estimated with cluster of households. Log(X) is natural logarithm of variable X; Std(X) is standard deviation of X; Corr(X,Y) is correlation between X and Y; Lab, Hou and Bus are, respectively, annual growth rates of labor income, home equity and business income; R_s is annual gross return of CRSP value-weighted market index; and R_f is annual gross return of the 30-day T-bill. "***", "**" and "*" denote statistical significance at the 1, 5 and 10 percent levels, respectively.

			Labor	Labor	
	Baseline	Labor Risk	& House Risk	& Business Risk	All Risks
_	(1)	(2)	(3)	(4)	(5)
Likelihood ratio test		(2)-(1)	(3)-(2)	(4)-(2)	(5)-(1)
Chi-square of likelihood ratio test		23.000	14.000	3.000	40.000
p-value of likelihood ratio test		(0.00)	(0.00)	(0.39)	(0.00)
Std(Lab)		-0.251***	-0.237***	-0.246***	-0.232***
		(-3.40)	(-3.19)	(-3.33)	(-3.12)
Corr(R _s , Lab)		-0.076***	-0.076***	-0.077***	-0.077***
		(-2.67)	(-2.69)	(-2.69)	(-2.71)
Corr(R _f , Lab)		0.069**	0.074**	0.069**	0.074**
		(2.33)	(2.50)	(2.33)	(2.49)
Std(Hou)			-0.089**		-0.090**
			(-2.10)		(-2.11)
Corr(R _s , Hou)			-0.031		-0.031
			(-1.00)		(-0.99)
Corr(R _f , Hou)			0.055*		0.054*
			(1.81)		(1.78)
Std(Bus)				-0.079*	-0.075*
				(-1.75)	(-1.67)
Corr(R _s , Bus)				0.024	0.022
				(0.40)	(0.37)
Corr(R _f , Bus)				0.049	0.053
				(0.86)	(0.92)
Corr(Lab, Hou)			-0.057**		-0.056**
			(-2.31)		(-2.28)
Corr(Lab, Bus)				0.013	0.017
				(0.20)	(0.25)
Corr(Hou, Bus)					0.016
					(0.35)
Head has a job	0.027	0.076**	0.075**	0.084**	0.081**
	(0.82)	(2.14)	(2.10)	(2.35)	(2.28)
Owns a house	0.595***	0.593***	0.646***	0.601***	0.654***
	(4.52)	(4.50)	(4.83)	(4.56)	(4.88)

Has a Business	-0.065***	-0.057***	-0.055***	-0.040**	-0.039**
	(-3.90)	(-3.45)	(-3.35)	(-2.02)	(-1.98)
Head in unemployment	-0.033	-0.027	-0.026	-0.027	-0.026
	(-1.36)	(-1.13)	(-1.07)	(-1.13)	(-1.06)
Head and wife in same industry	-0.045*	-0.043*	-0.042*	-0.042*	-0.041*
	(-1.93)	(-1.83)	(-1.83)	(-1.78)	(-1.77)
Log(Age)	0.805*	0.719	0.779	0.743	0.806*
	(1.69)	(1.50)	(1.63)	(1.55)	(1.68)
$(Log(Age))^2$	-0.114*	-0.102	-0.110*	-0.105*	-0.114*
	(-1.79)	(-1.61)	(-1.73)	(-1.65)	(-1.78)
Log(Family size)	-0.062***	-0.062***	-0.061***	-0.063***	-0.062***
	(-4.36)	(-4.35)	(-4.26)	(-4.40)	(-4.32)
Race-if white	0.213***	0.215***	0.216***	0.215***	0.215***
	(7.45)	(7.58)	(7.66)	(7.58)	(7.65)
High school	0.142***	0.140***	0.142***	0.140***	0.141***
	(6.39)	(6.33)	(6.42)	(6.29)	(6.38)
College	0.291***	0.287***	0.286***	0.286***	0.286***
	(11.9)	(11.7)	(11.7)	(11.7)	(11.7)
Log(Income)	0.085***	0.083***	0.083***	0.083***	0.084***
	(8.44)	(8.21)	(8.26)	(8.26)	(8.31)
Log(Wealth)	0.345***	0.345***	0.344***	0.346***	0.344***
	(28.7)	(28.7)	(28.5)	(28.7)	(28.5)
Log(House value)	-0.074***	-0.075***	-0.077***	-0.075***	-0.077***
	(-6.10)	(-6.13)	(-6.29)	(-6.18)	(-6.33)
Log(Unpaid Mortgage)	0.016***	0.015***	0.016***	0.016***	0.016***
	(9.43)	(9.33)	(9.63)	(9.37)	(9.67)
Log(Head Labor Income)	-0.003	-0.002	-0.002	-0.003	-0.003
	(-1.24)	(-1.06)	(-1.06)	(-1.44)	(-1.41)
Constant	Yes	Yes	Yes	Yes	Yes
Industry fixed effect	Yes	Yes	Yes	Yes	Yes
State fixed effect	Yes	Yes	Yes	Yes	Yes
Year fixed effect	Yes	Yes	Yes	Yes	Yes
Observations	21,938	21,938	21,938	21,938	21,938
Pseudo R-square	0.231	0.232	0.233	0.233	0.233

Table 5 Marginal Effects of Background Risk Factors on Stock Market Participation and Portfolio Holdings

Panel A reports the impacts of background risks on stock market participation and stock holding for the baseline model, while Panel B reports these impacts for different education groups. In each panel, Columns (1)-(3) report marginal effects of various background risks on stock market participation. Using estimated coefficients from the Logit regression (Table 3 Column 5), we assume that the corresponding risk factors change one standard deviation from their sample means while holding all other variables at their sample averages. Columns (4)-(6) report the marginal effects of background risks on stock holdings relative to financial wealth conditional on participation. Using estimated coefficients from the Tobit regression (Table 4 Column 5), we assume that the corresponding risk factors change one standard deviation from their sample means while holding all other variables at their sample means while holding all other variables at their sample means while holding all other variables at their sample means while holding all other variables at their sample means while holding all other variables at their sample means while holding all other variables at their sample averages. *Std(X)* is standard deviation of X; *Corr(X,Y)* is correlation between X and Y; *Lab*, *Hou* and *Bus* are, respectively, annual growth rates of labor income, home equity and business income; R_s is annual gross return of CRSP value-weighted market index; and R_f is annual gross return of the 30-day T-bill.

Panel A: Marginal Effect of Three Types of Background Risk Factors								
	Stock ma	rket participation (in	percent)	Stockholding re	Stockholding relative to financial wealth (in percent)			
	At sample means	At sample Increase one means std. dev. Change			Increase one std. dev.	Change		
	(1)	(2)	(3)	(4)	(5)	(6)		
Labor income risk Std(Lab), Corr(R _s , Lab), Corr(R _f , Lab)	34.21	27.95	-6.26	37.14	35.23	-1.92		
Housing risk Std(House), Corr(R _s , House), Corr(R _f , House)	34.21	30.80	-3.41	37.14	35.99	-1.16		
Business income risk Std(Bus), Corr(R _s , Bus), Corr(R _f , Bus)	34.21	32.58	-1.63	37.14	36.68	-0.47		
All risks 12 variables	34.21	23.39	-10.82	37.14	33.45	-3.69		
Panel B: Marginal Effect of 12 Background Risk Factors for Different Education Groups								

_	Stock ma	rket participation (in	percent)	Stockholding	Stockholding relative to financial wealth (in percent		
	At sample means	Increase one sample std. dev.	Change	At sample means	Increase one sample std. dev.	Change	
No high school	24.26	15.83	-8.43	32.71	29.63	-3.07	
High school	31.78	21.47	-10.31	36.51	32.91	-3.60	
College	44.85	32.32	-12.53	41.09	36.85	-4.24	

Table 6 Estimations Using Alternative Background Risk Measures

This table reports the results using backward rolling-over, forward-rolling over background risk measures, and the cross-sectional regressions. Std(X) is standard deviation of variable X; Corr(X,Y) is correlation between X and Y; Lab, Hou and *Bus* are, respectively, annual growth rates of labor income, home equity and business income; R_s is annual gross return of CRSP value-weighted market index; and R_f is annual gross return of the 30-day T-bill. The data are a 9-year unbalanced panel for 4,756 households for the years 1984, 1989, 1994, 1999, 2001, 2003, 2005, 2007, and 2009. "***", "**" and "*" denote statistical significance at the 1, 5 and 10 percent levels, respectively.

Different measures	Backward	looking	Forward 1	looking	Cross-sec	Cross-sectional	
Dependent variable	Participation	Portfolio	Participation	Portfolio	Participation	Portfolio	
Model	Logit	Tobit	Logit	Tobit	Logit	Tobit	
	(1)	(2)	(3)	(4)	(5)	(6)	
Std(Lab)	-0.791***	-0.162***	-0.867***	-0.192***	-1.098**	-0.110**	
	(-3.50)	(-2.76)	(-3.55)	(-2.89)	(-2.41)	(-2.26)	
Corr(R _s , Lab)	-0.146**	-0.036*	-0.077	0.003	-0.226	-0.046**	
	(-1.97)	(-1.86)	(-0.94)	(0.14)	(-1.30)	(-2.46)	
Corr(R _f , Lab)	0.089	0.025	0.176**	0.030	0.399**	0.052***	
	(1.18)	(1.27)	(1.98)	(1.20)	(2.25)	(2.73)	
Std(Hou)	-0.460***	-0.085**	-0.053	-0.025	-0.839***	-0.038	
	(-3.83)	(-2.57)	(-0.34)	(-0.58)	(-2.95)	(-1.24)	
Corr(R _s , Hou)	-0.064	-0.029	-0.082	-0.036	-0.186	-0.017	
	(-0.81)	(-1.39)	(-1.02)	(-1.63)	(-1.03)	(-0.83)	
Corr(R _f , Hou)	0.010	0.016	0.101	0.028	0.300	0.034*	
	(0.13)	(0.80)	(1.19)	(1.22)	(1.63)	(1.65)	
Std(Bus)	0.069	0.018	-0.340	-0.124**	-0.301	-0.011	
	(0.35)	(0.38)	(-1.35)	(-1.97)	(-0.78)	(-0.25)	
Corr(R _s , Bus)	0.383*	0.092*	-0.279	-0.123*	0.444	0.051	
	(1.73)	(1.76)	(-1.01)	(-1.76)	(0.91)	(0.92)	
Corr(R _f , Bus)	0.111	-0.008	0.107	0.075	0.463	0.061	
	(0.47)	(-0.16)	(0.39)	(1.13)	(1.02)	(1.20)	
Corr(Lab, Hou)	-0.090	-0.038*	-0.043	-0.040	-0.210	-0.041**	
	(-1.05)	(-1.68)	(-0.34)	(-1.20)	(-1.38)	(-2.47)	
Corr(Lab, Bus)	-1.089**	-0.143	0.211	0.026	-0.029	0.036	
	(-2.13)	(-1.26)	(0.21)	(0.10)	(-0.068)	(0.79)	
Corr(Hou, Bus)	0.446	0.057	0.313	0.049	0.161	0.008	
	(1.51)	(0.82)	(0.56)	(0.46)	(0.53)	(0.24)	
Other controls	Yes	Yes	Yes	Yes	Yes	Yes	
Year fixed effect	Yes	Yes	Yes	Yes	No	No	
State fixed effect	Yes	Yes	Yes	Yes	Yes	Yes	
Industry fixed effect	Yes	Yes	Yes	Yes	Yes	Yes	
Observations	19,298	19,298	9,336	9,343	4,716	4,726	
Pseudo R-square	0.266	0.229	0.256	0.224	0.326	0.478	

Table 7 Alternative Estimations of Portfolio Holdings

OLS regression is applied to a subsample of households which have participated in stock market in current and/or prior years. In the Heckman selection model which is estimated using a two-stage method, we include lagged participation and stockholding in the participation and portfolio holding equations, respectively. Only the second-stage results are reported. *PflStk_1 (PflStk_2, PflStk_3)* stands for the ratio of stockholdings to financial wealth (total wealth excluding home equity, total wealth including home equity). *Std(X)* is standard deviation of variable X; *Corr(X,Y)* is correlation between X and Y; *Lab, Hou* and *Bus* are, respectively, annual growth rates of labor income, home equity and business income; R_s is annual gross return of CRSP value-weighted market index; and R_f is annual gross return of the 30-day T-bill. The data are a 9-year unbalanced panel for 4,756 households for the years 1984, 1989, 1994, 1999, 2001, 2003, 2005, 2007, and 2009. "***", "**" and "*" denote statistical significance at the 1, 5 and 10 percent levels, respectively.

		Tobit			OLS			Heckman	
Dependent variable	PflStk_1	PflStk_2	PflStk_3	PflStk_1	PflStk_2	PflStk_3	PflStk_1	PflStk_2	PflStk_3
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Std(Lab)	-0.232***	-0.230***	-0.158***	0.024	-0.028	-0.022	0.03	-0.025	-0.043*
	(-3.12)	(-3.85)	(-3.94)	(0.68)	(-0.87)	(-0.96)	(0.79)	(-0.69)	(-1.74)
Corr(R _s , Lab)	-0.077***	-0.064***	-0.039**	-0.026**	-0.024**	-0.014*	-0.001	-0.01	0.003
	(-2.71)	(-2.81)	(-2.52)	(-1.97)	(-2.07)	(-1.69)	(-0.045)	(-0.68)	(0.32)
Corr(R _f , Lab)	0.074**	0.072***	0.047***	0.022	0.032**	0.022**	-0.004	0.023	0.006
	(2.49)	(2.96)	(2.84)	(1.55)	(2.48)	(2.37)	(-0.28)	(1.54)	(0.54)
Std(Hou)	-0.090**	-0.097***	-0.037	0.004	-0.028	0.011	0.026	-0.013	0.027*
	(-2.11)	(-2.77)	(-1.61)	(0.18)	(-1.30)	(0.75)	(1.17)	(-0.63)	(1.89)
Corr(R _s , Hou)	-0.031	-0.027	-0.015	-0.033**	-0.028*	-0.015	-0.035**	-0.029*	-0.017*
	(-0.99)	(-1.05)	(-0.93)	(-2.11)	(-1.88)	(-1.53)	(-2.20)	(-1.93)	(-1.67)
Corr(R _f , Hou)	0.054*	0.041	0.029*	0.025	0.02	0.016	0.032**	0.021	0.022**
	(1.78)	(1.63)	(1.80)	(1.61)	(1.37)	(1.63)	(2.02)	(1.38)	(2.13)
Std(Bus)	-0.075*	-0.076**	-0.058**	-0.024	-0.035	-0.032**	-0.019	-0.045*	-0.032**
	(-1.67)	(-2.06)	(-2.33)	(-1.01)	(-1.54)	(-2.04)	(-0.78)	(-1.96)	(-2.07)
Corr(R _s , Bus)	0.022	0.016	-0.004	0.022	0.013	-0.006	0.009	0.025	0.001
	(0.37)	(0.34)	(-0.13)	(0.64)	(0.40)	(-0.28)	(0.27)	(0.82)	(0.042)
Corr(R _f , Bus)	0.053	0.023	0.03	0.039	0.017	0.024	0.022	0.007	0.017
	(0.92)	(0.49)	(0.95)	(1.23)	(0.60)	(1.23)	(0.73)	(0.26)	(0.86)
Corr(Lab, Hou)	-0.056**	-0.045**	-0.025*	-0.014	-0.012	-0.004	-0.014	-0.021*	-0.004
	(-2.28)	(-2.23)	(-1.91)	(-1.10)	(-1.10)	(-0.48)	(-1.08)	(-1.73)	(-0.52)
Corr(Lab, Bus)	0.017	-0.002	-0.009	-0.012	-0.022	-0.016	-0.015	-0.016	-0.024
	(0.25)	(-0.028)	(-0.24)	(-0.36)	(-0.68)	(-0.71)	(-0.55)	(-0.61)	(-1.34)
Corr(Hou, Bus)	0.016	0.019	0.016	0.006	0.014	0.011	-0.008	0	-0.001
	(0.35)	(0.48)	(0.62)	(0.24)	(0.61)	(0.69)	(-0.35)	(-0.0083)	(-0.067)
Lagged PtfStk							0.321***	0.367***	0.363***
							(26.0)	(29.7)	(31.3)
Lamda							0.17	0.08	0.06
							(11.49)	(6.59)	(7.38)
Other Controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year fixed effect	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
State fixed effect	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Industry fixed effect	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Observations	21,938	21,938	21,938	13,976	13,976	13,976	17,771	17,771	17,771
R-square	0.23	0.28	0.38	0.25	0.25	0.26			

Figure 1 Cross-Sectional Dispersion of Background Risk Factors

This figure presents the cross-section distributions of 12 background risk variables. The sample includes observations for households that are exposed to the corresponding risks. The definitions of the variables are provided in Table 1. Grey bars represent the density, dash line is the kernel density curve, and solid line is the normal density curve.





Endnotes

⁴ This finding is different from Heaton and Lucas (2000b), who show that business income risk is more important than labor income risk. The discrepancy could be due to the fact that we use a sample of ordinary households, most of whom do not have a private business while they use the sample of wealthy households with a private business.

⁵ We consider labor income as dividends of the unobserved human capital, growth of home equity as returns on housing investment, and business income as dividends of business investment.

 6 We also use a two-sided Tobit model with the lower limit equal to 0 (a household holds no stocks) and the upper limit equal to 1 (a household holds only stocks). The results do not change significantly and are available upon request.

⁷ One challenge of the Heckman selection model is to find valid instrumental variables, which are related to stock market participation but not to stock holdings conditional on participation. However, in the empirical tests, we find that total wealth is significantly related to portfolio shares. Furthermore, education, which is regarded as a proxy for fixed cost for stock market participation is also significantly related to stock holdings. Hence, it is hard to identify good instrumental variables. Therefore, we report Tobit as the baseline model, and provide the Heckman and OLS models for robustness checks.

⁸ Petersen (2009) shows that given a large number of firms (households in our case), and a small number of years, correct standard errors can be obtained by including time dummies and then estimating standard errors with clustering by firms (households).

⁹The estimation of Logit and Tobit models is sensitive to the distributional assumptions about the error terms. In unreported robustness check, we also calculate nonparametric t-statistics using bootstrapped standard errors.

¹⁰ The PSID provides data on risk tolerance in the 1996 survey. Adding the risk tolerance variable in our regressions does not change our basic results. This test is not reported but is available upon request.

¹¹ The original PSID sample consisted of two independently selected samples: a cross-sectional national sample (the SRC sample) and a national sample of low-income families (the SEO sample). We exclude the SEO sample to generate a representative sample of the U.S. population. The PSID is designed to capture demographic and income dynamics of U.S. households over a long period. Households which were selected in the 1968 survey have been resurveyed thereafter. The split-off households (households established by children of the originally selected families) have been added to the sample each year.

¹² Most surveys are conducted in the springs and therefore income and wealth data are for the previous years.

¹³ Because the post-1999 stock holdings data may not be accurate, we conduct a robustness test using prior-1999 data and find similar results. These results are not reported but are available upon request.

¹⁴ In unreported robustness tests, we construct the background risk variables using data before 1997 when we have consecutive annual observations and obtain similar results.

¹⁵ Alternatively, we define the head of household business income as business income from assets only, and include business income from labor in the head of household labor income. Under this definition, none of our results change significantly. The results are available upon request.

¹⁶ We find that the standard deviation of the growth rate of self-reported market values of owner-occupied house has a negative impact on stock market participation and stock holdings even more significantly than the standard deviation of the growth rate of home equity, whereas the correlations of the growth rate of house value with asset returns do not have a significant impact.

¹⁷ To check for robustness, we require a household to have at least 10 years of growth rates to calculate the standard deviation and correlation statistics and obtain similar results.

¹⁸ These results are not tabulated, but are available upon request.

¹⁹ In the PSID survey, many households that claim owning a private business do not report their business income. Overall, although 16 percent of households indicate owning a private business, only 7.54 percent provide business income information.

¹ Heaton and Lucas (2000a), Campbell (2006), and Cochrane (2006) provide excellent reviews of this literature.

 $^{^{2}}$ As a robustness check, we also employ the Heckman selection model, and OLS regression for a sub-sample consisting of only stockholders.

³ This finding seems to suggest that background risks are more important for stock market participation than for stock holdings. Further results using the Heckman selection model and OLS regressions of stockholders confirm this finding.

²⁰ We also examine the correlations among time-invariant, forward-looking and backward looking background risk measures. We find that the time-invariant measure is more highly correlated with the backward measure than with the forward measure. The backward and forward measures of standard deviation variables are highly correlated. In contrast, the backward and forward measures of correlation variables are not closely associated. ²¹ The overall effect need not equal the sum of the separate effects due to the nonlinearity of the Logit model.